Author’s response to reviews

Title: Applying systematic review search methods to the grey literature: A case study examining guidelines for school-based breakfast programs in Canada

Authors:

Katelyn Godin (kmgodin@uwaterloo.ca)
Jackie Stapleton (jstapleton@uwaterloo.ca)
Sharon Kirkpatrick (sharon.kirkpatrick@uwaterloo.ca)
Rhona Hanning (rhanning@uwaterloo.ca)
Scott Leatherdale (sleatherdale@uwaterloo.ca)

Version: 3 Date: 25 Sep 2015

Author’s response to reviews:

September 25, 2015

Dear Dr. Mcdaid.

On behalf of my co-authors, I would like to thank you once again for your prompt review of our revised manuscript entitled “Applying systematic review search methods to the grey literature: A case study examining guidelines for school-based breakfast programs in Canada”.

We have reviewed your comments, revised the manuscript and table accordingly, and included our responses below (denoted with **). The corresponding changes can be seen in track changes within the revised files.

Thank you for considering this manuscript for publication in Systematic Reviews. Please do not hesitate to contact me with any questions or concerns.

Sincerely,

Katelyn Godin, PhD (Candidate)
School of Public Health and Health Systems
University of Waterloo
200 University Ave West, BMH 2315
Waterloo, ON Canada N2L 3G1
kmgodin@uwaterloo.ca
Editor's comments:

Page 11, Results - refer to Figure 1 at end of first sentence, as there does not appear to be any reference to this figure.

** Revised as requested.

Page 11 - refer to Table 2 at end of the sentence "The targeted web searches..."

** Revised as requested.

Please add the reference number of each of the documents to Table 2 otherwise readers have to check the bibliography to tie up which reference in the text refers to which document in Table 2. Apologies - I should have spotted this last time.

** Revised as requested. Thank you so much for your efficiency and detail in reviewing this manuscript. Your comments throughout the review process have been helpful indeed.